Author's response to reviews

Title: A fatal case of bupropion (Zyban) hepatotoxicity with autoimmune features: a case report

Authors:

Fawwaz Humayun (humayuf@hotmail.com)
Thomas M Shehab (shehabt@trinity-health.org)
Joseph A Tworek (itworek@yahoo.com)
Robert J Fontana (rfontana@med.umich.edu)

Version: 7 Date: 26 August 2007

Author's response to reviews: see over
Edits to Bupropion manuscript from Robert J. Fontana, MD  August 16, 2007.

My specific edits are as follows:

1. Abstract- Insert the word "hepatocellular" prior to jaundice at the end of the conclusion.

2. Page 6 of results- Change "2.0" to "20 days" in the top line.

3. Page 6 of results- Change line 10 to "A decision was made to forego the liver biopsy and restart the patient on prednisone 60 mg per day and he was discharged home."

4. Page 6 of results- Delete the sentence "Antifungal therapy was not started."

5. Page 7 conclusions, line 10- Change to "October 2004."

6. Page 7 conclusions, lines 14 to 16- Simplify to "... we feel that the patient's overall clinical course and previously reported cases of bupropion-induced liver injury presenting with autoimmune features support our conclusion that bupropion was the most likely inciting agent."

7. Page 8 of conclusions, line 7- Capitalize "Liver" when referring to DILIN

8. Page 10 of conclusions, line 7- Change to "..., it is our opinion that this case is the >>>>>"

9. Page 10 of conclusions, line 11- Delete the word "potentially"

10. Page 11, author's contributions- Change to "RJF participated in patient care and drafting of the manuscript."